Author’s response to reviews

Title: Ethical values supporting the disclosure of incidental and secondary findings in clinical genomic testing: a qualitative study

Authors:

Marlies Saelaert (Marlies.Saelaert@ugent.be)
Heidi Mertes (Heidi.Mertes@UGent.be)
Tania Moerenhout (Tania.Moerenhout@UGent.be)
Elfride De Baere (Elfride.DeBaere@UGent.be)
Ignaas Devisch (Ignaas.Devisch@UGent.be)

Version: 4 Date: 20 Jan 2020

Author’s response to reviews:

To

Ms. Janelle Coore
Editor
BMC Medical Ethics

METH-D-19-00168 – R4

“Ethical values supporting the disclosure of incidental and secondary findings in clinical genomic testing: a qualitative study”

Dear Ms. Coore,

We sincerely thank you for your feedback. In what follows, we respond to the suggestions that are marked in bold.

Editor Comments:

1. Please note that the Consent for publication statement refers to consent obtained to include any individual person’s data in any form (including individual details, images or videos). Therefore, please state "Not applicable" in this section as no personal information is provided in your manuscript.
We have adjusted this as requested (p. 27).

2. *As a reminder* please proofread and ensure that when you upload your revised submission it is in the final form for publication. Please remove any tracked changes or highlighting and include only a single clean copy of the manuscript. Should you wish to respond to these revision requests, please include the information in the designated input box only.

We have removed all tracked changes and uploaded a clean copy of the manuscript.

We are looking forward to your answer.

Sincerely,

Prof. Ignaas Devisch

Senior author

Marlies Saelaert

Doctoral fellow